Dear Editors:

After my initial submission last week, I noticed two small (but important) typographical errors in my manuscript.

1. p. 7, last paragraph, sentence should read, "Pfeiffer et al. describe two cases of sub-clinical factor VII and factor X deficiency associated with..." The phrase "and factor X deficiency" had accidentally gotten deleted in successive author revisions.

2. p. 8, continuation of same paragraph, "Fukushima et al. found approximately 50% factor VII activity in two..." "Factor V" was accidentally typed in the previous version, instead of "Factor VII."